Author's response to reviews

Title: Prevalence, Impacts And Medical Managements Of Premenstrual Syndrome Among Female Students: Cross-Sectional Study In College Of Health Science, Mekelle University, Mekelle, Northern Ethiopia

Authors:

Fikru W Tolossa (fikrewaf@gmail.com)
Mebratu L Bekele (mebleg@gmail.com)

Version: 2 Date: 8 October 2013

Author's response to reviews:

In the revised version, the number of words of the title, abstract, and manuscript are mentioned including the number of references.